Author's response to reviews

Title: Factors associated with pastoral community knowledge and occurrence of mycobacterial infections in Human-Animal Interface areas of Nakasongola and Mubende districts, Uganda

Authors:

Clovice Kankya (clokankya@yahoo.com)
Adrian Muwonge (adrian.muwonge@nvh.no)
Susan Olet (susan.olet@ndsu.edu)
Musso Munyeme (munyeme@yahoo.co.uk)
Demelash Biffa (Aredares@veths.no)
John Opuda-Asibo (opas_jn@yahoo.com)
Eystein Skjerve (Eystein.Skjerve@nvh.no)
James Oloya (joloya2001@yahoo.co.uk)

Version: 9 Date: 9 August 2010

Author's response to reviews: see over
Cover letter to the Editor BMC Public Health series Editorial Production team

Journal: BMC Public Health

Title: Factors associated with pastoral community knowledge and occurrence of mycobacterial infections in Human-Animal Interface areas of Nakasongola and Mubende districts, Uganda

MS ID : 7809727993543937

Final Confirmation on the Manuscript:

I apologize for this delay in response. I had travelled to the countryside of Western Uganda.

I have made few typographical changes and minor editions. I confirm therefore that this is the final version of the Manuscript (Well read and edited).

Kindly advise me in case there is any other change to make on the Manuscript. I will be grateful for your response.

Looking forward to hearing from you

Thank you